Title: The association of hypertriglyceridemia with cardiovascular events and pancreatitis: A systematic review and meta-analysis

Authors:

- M. Hassan Murad (murad.mohammad@mayo.edu)
- Ahmad Hazem (ahazem@yahoo.dom)
- Fernando Coto-Yglesias (cotoyglesias@yahoo.com)
- Svitlana Dzyubak (sdzyubak@yahoo.com)
- Shabnum Gupta (shabnum.gupta@gmail.com)
- Irina Bancos (bancos.irina@mayo.edu)
- Melanie Lane (melanielane7@gmail.com)
- Patricia Erwin (erwin.patricia@mayo.edu)
- Lars Berglund (lars.berglund@ucdmc.ucdavis.edu)
- Tarig Elraiyah (elraiyah.tarig@mayo.edu)
- Victor Montori (montori.victor@mayo.edu)

Version: 3 Date: 21 March 2012

Author's response to reviews: see over
The association of hypertriglyceridemia with cardiovascular events and pancreatitis: A systematic review and meta-analysis

Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.

Competing interest section added.

Please include an Authors' contributions section before the Acknowledgments and Reference list.

Authors' contributions section added.

All contributors who do not meet the criteria for authorship should be listed in an acknowledgments section. Examples of those who might be acknowledged include a person who provided purely technical help, writing assistance, or a department chair who provided only general support. Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgments’ section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

Acknowledgements section added.